Author's response to reviews

Title: Prevalence of sexually transmitted infections among pregnant women with known HIV status in northern Tanzania.

Authors:

Sia E Msuya (siamsuya@hotmail.com)
Jacqueline Uriyo (jakeeuryi@yahoo.com)
Akhtar Hussain (akhtar.hussain@medisin.uio.no)
Elizabeth M Mbizvo (embizvo@mweb.co.zw)
Stig Jeansson (s.l.jeansson@medisin.uio.no)
Noel E Sam (noelsam@komc.ac.tz)
Babill Stray-Pedersen (babill.stray-pedersen@medisin.uio.no)

Version: 6 Date: 13 January 2009

Author's response to reviews:

All the tables have been made using tables tool i.e. the tables are divided into cells as requested.